Six versus 12 months' adjuvant trastuzumab in patients with HER2-positive early breast cancer: the PERSEPHONE non-inferiority RCT

Helena Earl,^{1,2,3*} Louise Hiller,⁴ Anne-Laure Vallier,⁵ Shrushma Loi,⁴ Karen McAdam,^{6,7} Luke Hughes-Davies,^{1,7} Daniel Rea,⁸ Donna Howe,⁴ Kerry Raynes,⁴ Helen B Higgins,⁴ Maggie Wilcox,⁹ Chris Plummer,^{10,11} Betania Mahler-Araujo,^{12,13} Elena Provenzano,^{3,12} Anita Chhabra,¹⁴ Sophie Gasson,⁴ Claire Balmer,⁴ Jean E Abraham,^{1,2,3} Carlos Caldas,^{1,2,3,15} Peter Hall,¹⁶ Bethany Shinkins,¹⁷ Christopher McCabe,¹⁸ Claire Hulme,^{17,19} David Miles,²⁰ Andrew M Wardley,^{21,22} David A Cameron¹⁶ and Janet A Dunn⁴ on behalf of the PERSEPHONE Steering Committee and Trial Investigators

¹Department of Oncology, University of Cambridge, Addenbrooke's Hospital, Cambridge, UK

- ²Cambridge Breast Cancer Research Unit, Cambridge University Hospitals NHS Foundation Trust, Cambridge, UK
- ³NIHR Cambridge Biomedical Research Centre, Cambridge, UK
- ⁴Warwick Clinical Trials Unit, University of Warwick, Coventry, UK
- ⁵Cambridge Clinical Trials Unit Cancer Theme, Cambridge University Hospitals NHS Foundation Trust, Cambridge, UK
- ⁶Department of Oncology, North West Anglia NHS Foundation Trust, Peterborough City Hospital, Peterborough, UK
- ⁷Department of Oncology, Cambridge University Hospitals NHS Foundation Trust, Cambridge, UK
- ⁸Cancer Research UK Clinical Trials Unit (CRCTU), Institute of Cancer and Genomic Sciences, University of Birmingham, Birmingham, UK
- ⁹Independent Cancer Patients' Voice, London, UK
- ¹⁰Translational and Clinical Research Institute, Faculty of Medical Sciences, Newcastle University, Newcastle upon Tyne, UK
- ¹¹Department of Cardiology, Freeman Hospital, Newcastle upon Tyne Hospitals NHS Foundation Trust, Newcastle upon Tyne, UK
- ¹²Department of Histopathology, Cambridge University Hospitals NHS Foundation Trust, Cambridge, UK
- ¹³Metabolic Research Laboratories, University of Cambridge, Cambridge, UK

- ¹⁴Pharmacy, Cambridge University Hospitals NHS Foundation Trust, Addenbrooke's Hospital, Cambridge, UK
- ¹⁵Cancer Research UK Cambridge Institute, University of Cambridge, Cambridge, UK ¹⁶Edinburgh University Cancer Research Centre, Institute of Genetics and Molecular
 - Medicine, University of Edinburgh, Western General Hospital, Edinburgh, UK
- ¹⁷Academic Unit of Health Economics, University of Leeds, Leeds, UK
- ¹⁸Institute of Health Economics, Edmonton, AB, Canada
- ¹⁹Health Economics Group, University of Exeter Medical School, Exeter, UK
- ²⁰Medical Oncology, Mount Vernon Cancer Centre, Northwood, UK
- ²¹NIHR Manchester Clinical Research Facility at The Christie NHS Foundation Trust, Manchester Academic Health Science Centre, Manchester, UK
- ²²Division of Cancer Sciences, Faculty of Biology, Medicine and Health, University of Manchester, Manchester Academic Health Science Centre, Manchester, UK

*Corresponding author hme22@cam.ac.uk

Declared competing interests of authors: Helena Earl reports grants from Roche (Basel, Switzerland) and Sanofi-Aventis (Paris, France), personal fees and travel expenses from Daiichi Sankyo (Tokyo, Japan), AstraZeneca plc (Cambridge, UK) and Intas Pharmaceuticals (Ahmedabad, India), travel expenses from Pfizer Inc. (New York, NY, USA) and Amgen Inc. (Thousand Oaks, CA, USA) and personal fees from prIME Oncology (Atlanta, GA, USA), all outside the submitted work. Karen McAdam reports grants from Roche and personal fees from Roche, Novartis International AG (Basel, Switzerland), Pfizer and Eisai Co., Ltd (Tokyo, Japan), all outside the submitted work. Daniel Rea reports personal fees and grants from Roche during the conduct of the study, as well as personal fees from Novartis, Pfizer, Genomic Health (Redwood City, CA, USA) and Daiichi Sankyo, and grants from Celgene Corporation (Summit, NJ, USA), all outside the submitted work. Chris Plummer reports personal fees and non-financial support from Roche Products Limited, Novartis UK Limited, Pfizer UK Limited, Celgene and Incyte Corporation (Wilmington, DE, USA) for attending education meetings. He also reports personal fees and non-financial support from Amgen Limited for attending education meetings and advisory boards, all outside the submitted work. Jean Abraham reports fees to her institution, and accommodation and travel expenses from AstraZeneca for session boards and advisory chairs, as well as fees to her institution, and accommodation and travel expenses from Pfizer for a lecture, all outside the submitted work. Carlos Caldas reports grants from Genentech, Inc. (South San Francisco, CA, USA), Roche, Servier Laboratories (Suresnes, France) and AstraZeneca outside the submitted work, and that he is a member of the AstraZeneca iMED External Science Panel. Peter Hall reports grants from Roche, Pfizer, AstraZeneca, Novartis, Eisai and Daiichi Sankyo outside the submitted work. Christopher McCabe's institution holds research contracts with Roche and reports grants from Roche, all outside the submitted work. David Miles reports personal fees from Roche/Genetech, outside the submitted work. Claire Hulme reports that she is a member of the National Institute for Health Research (NIHR) Health Technology Assessment Commissioning Board. Andrew M Wardley reports personal fees from Roche, Napp Pharmaceuticals Ltd (Cambridge, UK), Amgen, Merck Sharp & Dohme (Hoddesdon, UK), Novartis, Pfizer, AstraZeneca, Laboratoires Pierre Fabre (Paris, France), Accord (Barnstaple, UK), Athenex (Buffalo, NY, USA), Gerson Lehrman Group (New York, NY, USA), Coleman Research Expert Network Group (New York, NY, USA) and Guidepoint Global (New York, NY, USA). He also reports personal fees and other from Eli Lilly and Company (Indianapolis, IN, USA) and Daiichi Sankyo, all outside the submitted work. He is leading the National Cancer Research Institute Breast Group Initiative to develop the next de-escalation trial for HER2-positive breast cancer. David A Cameron reports funds to his institution from Novartis, Astrazeneca, Pfizer, Roche, Eli Lilly and Company, Puma Biotechnology (Los Angeles, CA, USA), Daiichi Sankyo, Synthon (Nijmegen, the Netherlands), SeaGen International GmbH (Zug, Switzerland), Zymeworks (Vancouver, BC, Canada), Elsevier (Amsterdam, the Netherlands), European Cancer Organisation (Brussels, Belgium), Celgene Corporation, Succinct Medical Communications (Wilmington, DE, USA), Immutep (Sydney, NSW, Australia),

Oncolytics Biotech (U.S) Inc. (San Diego, CA, USA), Celldex Therapeutics Inc. (Hampton, NJ, USA), San Antonio Breast Cancer Consortium (TX, USA), Highfield Communication (Oxford, UK), Samsung Bioepis Co. Ltd (Incheon, South Korea), prIME Oncology, Merck Sharp & Dohme Ltd, Prima Biomed Ltd, RTI Health Solutions (Research Triangle, NC, USA) and Eisai, all outside the submitted work. Janet A Dunn reports that she is a member of the NIHR Efficacy and Mechanism Evaluation funding board and an NIHR senior investigator.

Published August 2020 DOI: 10.3310/hta24400

Plain English summary

The PERSEPHONE RCT Health Technology Assessment 2020; Vol. 24: No. 40 DOI: 10.3310/hta24400

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

The background

There are several different types of breast cancer and some are called 'HER2 positive'. These cancers can often be cured by treatment with chemotherapy and a drug called trastuzumab (also known as Herceptin[®]; Roche, Basel, Switzerland). Although the first trials of trastuzumab used 12 months treatment, we did not know if less treatment could work as well. A small trial in Finland showed that giving trastuzumab for just 9 weeks was also effective. We know that trastuzumab can have some side effects, including heart problems, so it was important to see if we could reduce the length of treatment time, which is usually 12 months.

What did we do?

We wanted to find out if we could treat patients safely with 6 months rather than 12 months of trastuzumab. We carried out a clinical trial called PERSEPHONE, in which over 4000 patients with this type of early breast cancer took part. Half of the patients were given 12 months of trastuzumab and half were given 6 months of trastuzumab.

What did we find?

We found that the two groups of patients had very similar benefit from treatment. At 4 years after diagnosis 90.3% of those who had received 12 months of trastuzumab were alive and free of any breast cancer recurrence, compared with 89.5% of those who had received 6 months. In other words, 125 patients would need to be treated with 12 months' trastuzumab rather than 6 months' trastuzumab for one more person to be alive and cancer-free 4 years from diagnosis.

The side effects?

Severe side effects of trastuzumab were seen on at least one occasion in 24% of 12-month patients compared with 19% of 6-month patients. More patients receiving 12 months of trastuzumab had to stop trastuzumab early because of heart problems (8% of 12-month patients compared with 3% of 6-month patients).

What does this all mean?

We have shown that 6 months of trastuzumab has similar outcomes to 12 months in treating patients with HER2-positive early breast cancer but with fewer severe side effects, including heart problems, fewer visits to hospital for patients and significant cost savings for the NHS.

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 3.370

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, the Cochrane Library and Clarivate Analytics Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 06/303/98. The contractual start date was in April 2007. The draft report began editorial review in July 2019 and was accepted for publication in January 2020. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2020. This work was produced by Earl *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

Editor-in-Chief of Health Technology Assessment and NIHR Journals Library

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Senior Clinical Researcher, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk